Dear Editor

We have addressed the revisions requested using tracking in the manuscript submitted as word document.

1. Formatting

In order to be in line with journal requirements, we have included the following headings within the abstract and manuscript: Background, Methods, Results, Discussion, Conclusions.

2. Abbreviations

A list of all the abbreviations used in the manuscript has been placed just before the Declarations section. All abbreviations are still be defined in the text at first use.

3. Consent for publication

We have changed this to “Not Applicable” in this section.
4. Authors' Contributions

We have included the individual contributions of each author, using their full initials, not their full name, in the Authors’ Contributions section. We have also included a statement in the Authors' contributions section to the effect that all authors have read and approved the final manuscript, and have emails confirming this form each author.

Thank you

Kind regards

Kris

A/Professor Kristine Barlow-Stewart

Faculty of Medicine and Health

The University of Sydney